Author's response to reviews

Title: Pattern and determinants of hospitalization during heat waves: an ecologic study

Authors:

Giuseppe Mastrangelo (giuseppe.mastrangelo@unipd.it)
Ugo Fedeli (fedeli@ulssasolo.ven.it)
Cristiana Visentin (visentin@ulssasolo.ven.it)
Giovanni Milan (milang@ulssasolo.ven.it)
Emanuela Fadda (emanuela.fadda@unipd.it)
Paolo Spolaore (spolaore@ulssasolo.ven.it)

Version: 7 Date: 16 July 2007

Author's response to reviews:

Dear Editor,

according to your suggestions we have performed the following changes:

we've moved the website citation from the text to the reference list, with reference numbers updated accordingly both in the text and the References section;

we've modified the Author's contribution according to the journal's format, and moved the section before Acknowledgements;

we've moved Figures titles from the images to the manuscript (after the reference list);

the Figures are now powerpoint files adequately cropped.

Best regards,

Giuseppe Mastrangelo